Author's response to reviews

Title: A sensitive flow cytometric methodology for studying the binding of L. chagasi to canine peritoneal macrophages

Authors:

Ricardo RG Goncalves (pepe@icb.ufmg.br)
Etel Rocha ERV Vieira (etelvieira@terra.com.br)
Maria Norma MNM Melo (melo@icb.ufmg.br)
Kenneth John KJG Gollob (kigollob@icb.ufmg.br)
David DM Mosser (dm268@umail.umd.edu)
Wagner Luiz WLT Tafuri (wagner@icb.ufmg.br)

Version: 5 Date: 4 May 2005

Author's response to reviews:

Dear assistant editor, Iratxe Puebla,
We are sending the correct version of manuscript.
We do apologize about the mistakes.
Please contact us if you have any problems or questions.
Thank you very much,
Ricardo Goncalves